

# Association between age and severity to leptospirosis in children.

Gilles Guerrier, Pauline Hie, Ann-Claire Gourinat, Emilie Huguon, Yann Polfrit, Cyrille Goarant, Eric d'Ortenzio, Isabelle Missotte

### ▶ To cite this version:

Gilles Guerrier, Pauline Hie, Ann-Claire Gourinat, Emilie Huguon, Yann Polfrit, et al.. Association between age and severity to leptospirosis in children.. PLoS Neglected Tropical Diseases, 2013, 7 (9), pp.e2436. 10.1371/journal.pntd.0002436 . pasteur-00983259

### HAL Id: pasteur-00983259 https://riip.hal.science/pasteur-00983259

Submitted on 25 Apr 2014  $\,$ 

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

# Association between Age and Severity to Leptospirosis in Children

## Gilles Guerrier<sup>1\*</sup>, Pauline Hie<sup>2</sup>, Ann-Claire Gourinat<sup>3</sup>, Emilie Huguon<sup>2</sup>, Yann Polfrit<sup>2</sup>, Cyrille Goarant<sup>4</sup>, Eric D'Ortenzio<sup>5</sup>, Isabelle Missotte<sup>2</sup>

1 Intensive Care Unit, Centre Hospitalier Territorial, Noumea, New Caledonia, 2 Paediatrics, Centre Hospitalier Territorial, Noumea, New Caledonia, 3 Laboratory of Virology, Institut Pasteur in New Caledonia, Noumea, New Caledonia, 4 Laboratory of Microbiology Research Unit, Institut Pasteur in New Caledonia, Noumea, New Caledonia, 5 Infectious Diseases Epidemiology Unit, Institut Pasteur in New Caledonia, Noumea, New Caledonia, Noumea, New Caledonia, 5 Infectious Diseases Epidemiology Unit, Institut Pasteur in New Caledonia, Noumea, New Caledonia, Noumea, New Caledonia, Sourcea, New Caledonia, Nourcea, New Caledonia, Sourcea, New Caledonia, Nourcea, New Caledonia, Nour

#### Abstract

**Background:** In endemic areas, leptospirosis is more common and more severe in adults compared with children. Reasons to explain this discrepancy remain unclear and limited data focusing on adolescents are available. The objective of the study was to describe disease spectrum and outcome differences in children and adolescents admitted for leptospirosis in a large at-risk population.

Methods: Clinical and laboratory data were obtained on hospitalized cases in New Caledonia from 2006 to 2012.

**Results:** Data of 60 patients <18 years of age (25 children under 14 and 35 adolescents aged 14 to 17) with confirmed leptospirosis were analyzed. Compared with children, adolescents presented more often with classic features of Weil disease (p = 0.02), combining hepatic and renal involvement with or without pulmonary participation. Jarisch-Herxheimer reactions were observed more often among adolescents (p < 0.01). The overall case fatality rate was low (1 adolescent versus 0 children).

*Conclusion:* Severe leptospirosis in adolescents may be more likely to show adults' characteristics compared with children. Further studies are required to explore age-dependant host factors, including puberty-related physiological changes.

Citation: Guerrier G, Hie P, Gourinat A-C, Huguon E, Polfrit Y, et al. (2013) Association between Age and Severity to Leptospirosis in Children. PLoS Negl Trop Dis 7(9): e2436. doi:10.1371/journal.pntd.0002436

Editor: Jessica N. Ricaldi, Universidad Peruana Cayetano Heredia, Peru

Received May 16, 2013; Accepted August 9, 2013; Published September 26, 2013

**Copyright:** © 2013 Guerrier et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The authors received no specific funding for this study.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: guerriergilles@gmail.com

#### Introduction

Leptospirosis is an important zoonosis of worldwide distribution caused by pathogenic spirochaetes of the genus Leptospira. Humans usually become infected through contact with water or soil contaminated by the urine of mammalian reservoirs such as rodents, dogs, cattle and pigs [1].

Infection most commonly results in asymptomatic or selfresolving illness both in adults and children [2]. In severe cases requiring hospitalization, the disease is however potentially fatal classically presenting with jaundice and renal dysfunction (Weil disease) with or without pulmonary hemorrhagic manifestations. According to estimates from the World Health Organization, more than 500,000 severe cases occur every year worldwide, mostly in tropical and sub-tropical regions.

Along with experienced clinicians' beliefs, several studies suggest leptospirosis to produce more severe presentation in adults compared with children [3,4,5,6,7]. Pathogen as well as hostrelated factors are believed to play a role in the development of severe leptospirosis in adults [1,8]. However, factors responsible for the milder presentation among children remain unclear. There is limited information available about symptomatic leptospirosis in the under 18 age group in the Pacific region. New Caledonia is an overseas French-administered territory located in the South Pacific and located 1500 kilometres East of Australia. According to the 2009 census, the population is 245 580 inhabitants with a mean increase of 1.7% per year since 1996 [9]. The population has good access to the health care system of European standards. Climate in New Caledonia is marked by a cool and dry season (from June to September) and a warm and wet one (from December to March). Leptospirosis is endemic to New Caledonia, and is a leading cause of hospital admission during the rainy season [10,11]. From 2006 to 2009, the average annual incidence was 45 cases per 100,000 inhabitants but reached 150 per 100,000 inhabitants during the rainiest months.

The objective of the study was to describe disease spectrum and outcome differences in children and adolescents admitted for laboratory-confirmed leptospirosis in a large at-risk population. We hypothesized that adolescents were more likely to present with all the classic features of Weil disease compared with younger children.

#### Methods

#### Study design

We carried out an observational retrospective study among patients aged under 18 years old with a biologically confirmed

#### **Author Summary**

Leptospirosis is endemic in tropical areas and seems to affect adults more often and more severely than children. Factors responsible for such differences have not been clearly established. However, host-related factors are believed to play a role in the development of severe leptospirosis. The study aimed to describe disease spectrum and outcome differences in confirmed cases in children and adolescents in New Caledonia. One major finding is the milder presentation of children compared with adolescents. Clinical and biological characteristics in adolescents are similar to adults, including occurrence of Jarisch-Herxheimer reactions. Further studies are required to explore age-dependant host factors, including pubertyrelated physiological changes.

leptospirosis admitted between January 2006 and December 2012 in either of two public hospitals (Centre Hospitalier Territorial, Noumea and Centre Hospitalier du Nord, Koumac). Demographic, epidemiologic, clinical, and laboratory information were recorded. Outpatients testing results for leptospirosis or adults over 18 were not included in the analysis. Leptospirosis was defined by a compatible clinical syndrome (any combination of fever, chills, myalgia, jaundice, conjunctival suffusion, renal failure, hemorrhage, or pulmonary failure) and laboratory confirmation with one or more of the following features: 1) positive results for a microscopic agglutination test (using a panel of 11 serovars) and one acute-phase serum sample titer >1:800, 2)seroconversion between times of testing acute-phase and convalescent phase serum samples, 3) a four-fold increase in titers between two examinations, or 4) a positive PCR. Biological tests performed at the Institut Pasteur in New Caledonia (IPNC) have been described in a previous study [8].

Patients were stratified by age group: 0-13 years of age (children) and >13 years (adolescents). Oliguria was defined as urine output <0.5 mL/kg/hour and pulmonary involvement was defined as dyspnea, rales, and/or chest radiographic abnormalities. Laboratory results refer to samples collected at the time of admission. Reference values were serum creatinine  $= 30-88 \ \mu g/$ dL, total bilirubin = 0.8-1.2 mg/dL, and platelet counts = 150 000-400 000/mm3. The time (in days) between onset of symptoms and hospitalization was compared between groups. Occurrence of Jarisch-Herxheimer reaction (JHR) was also noted when reported. JHR was defined as the combination of sudden onset of shivering or rigors, with rise in temperature, with or without a fall or a rise in blood pressure, increase of respiratory rate occurring after administration of the first dose of antibiotics.

#### Procedure and data collection

For each study participant, a standardized form was retrospectively completed. Clinical manifestations and medical history were collected as mentioned in medical records. Demographic data and laboratory results were extracted from electronic records. Both methods (MAT and PCR product sequence polymorphism) were used to identify the serogroup or the putative serogroup of the infecting strain.

#### Ethics statement

The study was approved by the Institutional Review Board of Centre Hospitalier Territorial. Informed consents were not obtained from the patients as this was a retrospective study. All data were anonymized.

#### Statistical analysis

STATA (Stat Corp., College Station, TX) was used for analysis. Quantitative and qualitative variables were compared by using paired Student's t-tests and chi-square tests, respectively. When the frequency of events was <5 or values did not follow normal distributions, Fisher's Exact and Mann-Whitney tests were used.

#### Results

A total of 128 patients with leptospirosis were diagnosed during the study period. Sixty eight of these cases were excluded from further analysis because they were not hospitalized in one of the two participating centres (n = 50) or records could not be traced (n = 16).

Of the 60 patients included in the study, 25 (42%) were children and 35 (58%) were adolescents. The majority of study subjects were boys (79% in children vs 77% in adolescents, p = 0.6) with a mean age of 9 [IQR 6–11] years for children and a mean age of 16 [IQR 14–17] for adolescents. Subjects were mostly Melanesian (83%) living in tribes in rural areas (85%) (Table 1).

The frequency of symptoms and complications are presented in Table 2 and biological parameters are presented in Table 3. Fever and jaundice were more frequent among adolescents whereas the incidence of conjunctival suffusion, myalgia, abdominal pain, headache and oliguria were not significantly different between groups. Mean total serum bilirubin and serum creatinine levels were higher in adolescent than paediatric groups (41 versus 7; p<0.001 and 108 versus 59; p<0.001, respectively). Platelet count was lower in adolescents (148 000 [95%CI 124 000–173 000]) than in children (202 000 [95%CI 157 000–248 000]) (p = 0.02). Compared with children, adolescents presented more often with classic features of Weil disease (p = 0.02), combining hepatic and renal involvement with or without pulmonary participation. The groups showed no differences with respect to pulmonary involvement.

Exposition factors for leptospirosis transmission did not differ between the groups; most patients in both groups had self-reported direct contact with water (swimming in rivers or canals, wading through water) (87% in children and 81% in adolescents), and direct or indirect contact with mammalian carriers (25% in children and 28% in adolescents).

The median time between onset of symptoms and initiation of antibiotics was 2 days for both groups. Antibiotics were administered intravenously for 5–7 days and included ampicillin (100 mg/kg of body weight/day) in 43 patients (18 children and 25 adolescents) and cefotaxime (100 mg/kg of body weight/day) in 17 patients (7 children and 10 adolescents).Jarisch-Herxheimer

**Table 1.** Demographic characteristics and medical histories

 of 60 paediatric leptospirosis in New Caledonia, 2006.

	Children	Adolescents	
Characteristic	n (%)	n (%)	p-value
Gender			
Male	18 (72)	27 (77)	0.6
Female	7 (28)	8 (23)	
Age (Mean [25–75 IQR])	9 [6–11]	16 [14–17]	
Ethnic group			
Melanesian	19 (76)	31 (88)	0.3
Other community	6 (24)	4 (12)	

doi:10.1371/journal.pntd.0002436.t001

 Table 2. Clinical symptoms in 60 leptospirosis children in New Caledonia, 2006–2012.

Clinical presentation at admission	Children n (%)	Adolescents n (%)	p-value
Fever (>38°) at admission	10 (40)	23 (66)	0.04
Jaundice	1 (4)	10 (28)	0.03
Headache	16 (64)	27 (77)	0.3
Abdominal pain	11 (44)	22 (63)	0.1
Myalgia	18 (72)	25 (71)	0.9
Conjonctival suffusion	4 (16)	9 (26)	0.4
Oliguria	2 (8)	7 (20)	0.2
Pulmonary involvement (cough, dyspnea, auscultation abnormalities)	7 (28)	11 (31)	0.6
Weil disease	0 (0)	7 (20)	0.02
Complications			
Alveolar hemorrhage	1 (4)	5 (14)	0.2
Myocarditis	0 (0)	1 (3)	0.5
Case-fatality rate	0 (0)	1 (3)	0.5

doi:10.1371/journal.pntd.0002436.t002

reactions were observed more often among adolescents (p < 0.01) (Table 4). The overall case fatality rate was 1.6% (1 adolescent versus 0 children). The single patient with fatal outcome required dialysis for acute renal failure, mechanical ventilation for alveolar hemorrhage and vasoactive drug for shock. Four other adolescents required vasoactive drugs only. No other patients required dialysis or mechanical ventilation.

Among the 26 cases for whom the serogroup was identified, 15 (58%) were Icterohaemorrhagiae (7/11 in children, 8/15 in adolescents). Other detected serogroups included Australis (n = 4), Pyrogenes (n = 4), Canicola (n = 2), and Panama (n = 1) (Table 3). For the 34 remaining cases, identification of the serogroup was not possible. At the time of admission, 8 patients had acute co-infections (5 children and 3 adolescents) with viral diseases (rotavirus n = 2; viral respiratory syncitial n = 1), or bacterial diseases (bacteraemia with Serratia n = 1; urinary tract infection n = 2, pyodermittis n = 1, tuberculosis n = 1). All patients diagnosed with coinfections were leptospirosis confirmed cases by PCR.

#### Discussion

This retrospective study allowed us to identify an age-dependant association with severity of leptospirosis. Similar observations in children have been reported in other settings [3,12]. The frequency of several classic severe disease manifestations were significantly lower among small children in this study compared with adolescents. Our study found that a substantial proportion of

**Table 3.** Initial laboratory findings among 60 leptospirosischildren in New Caledonia, 2006–2012.

Children		Adolescents		
Parameter*	n = 25	n = 35	p-value	
Bilirubine (mg/dL)	7 [5–12]	41 [34–55]	< 0.001	
Creatinine (µM/dL)	59 [62–65]	108 [85–132]	< 0.001	
Platelet count (G/L)	202 [157–248]	148 [124–173]	0.02	

\*mean [95%CI].

doi:10.1371/journal.pntd.0002436.t003

hospitalized children with leptospirosis had fewer of the classic features of Weil disease than adolescents. Although severe disease caused by leptospirosis may occur in the paediatric age group, clinical and biological presentation in adolescents overlaps with the spectrum seen in adults. Since most studies performed in children did not discriminate age groups, our findings are uneasy to compare with other series. However, some characteristics of small children in our study were similar to what has been previously reported in Thailand [13] and Brazil [7]. In contrast, our results differ from a previous report of 43 children, 4-14 years of age [14] showing more severe manifestations of leptospirosis than the present study, including renal failure and thrombocytopenia. Several factors may explain these differences: first, different leptospiral serogoups were identified in New Caledonia; second, time to refer was potentially higher in the Brazilian study; finally, host factors and higher organism loads may lead to more severe manifestations. Conversely, presentation of severe illness in adolescents was similar to clinical and biological profile reported in adults in Brazil [3] and New Caledonia [8].

Consistent with some previous studies [5,14], our results showed that overall case-fatality rates in children are lower than in adults. Recently identified factors associated with severe disease in adults in the same setting included tobacco use, leptospiral serogoup and delay between onset of symptoms and initiation of antibacterial therapy [8]. The discrepancy between children and adolescents does not appear related to differences in seroreactivity to leptospiral serogroups, which was similar in both groups. Similarly, onset of symptoms to administration of antibiotics was similar in both groups suggesting identical time to refer for children and adolescents. Host factors may contribute to this association and could include higher organism loads with increasing age. However, since risk factors for exposure were identical in children and adolescents, it is unlikely that initial bacterial inoculum was different in both groups. Age-dependant changes in innate and adaptative immune responses to leptospiral infection are plausible explanations for differences between children and adolescents. Several hypotheses regarding leptospirosis severity are based on host genetic susceptibility factors [15,16] and/or on bacterial virulence [17], although the virulence mechanisms are poorly understood and are probably multifactorial [18].

Table 4. Characteristics of infection and effects of antibiotics among 60 leptospirosis children in New Caledonia, 2006–2012.

	Children n (%)	Adolescents n (%)	p-value
Infection serogroup			
L. interrogans serogroup Icterohaemorrhagiae	7 (64)	8 (53)	0.7
Others	4 (36)	15 (47)	
Duration of symptoms before admission to hospital (days)*	2 (1–3)	2 (1–3)	0.2
Jarsich-Herxheimer reaction	1 (4)	12 (34)	0.005

\*mean [95%CI].

doi:10.1371/journal.pntd.0002436.t004

The male predominance reported in our study both in children and adolescents is very similar to previous studies of symptomatic leptospirosis [3,14]. Gender-specific activities may explain such differences. However, it is common in New Caledonia to see boys and girls swimming in water and assisting their parents with cultivating fields. The number of symptomatic cases admitted in paediatrics is low compared with adults over the same period of time [8]. This result is surprising since clinicians have a lower threshold to admit children with suspected or confirmed leptospirosis to the hospital. Age specific activities may explain this finding.

Another major finding of our study showed that JHRs were more likely to occur in adolescents. Although this adverse event is scarcely described in leptospirosis [19], our results support the presentation and outcome overlap between adolescents and adults. The incidence of JHR in children included in our study is in line with the few studies reporting JHR in paediatric cases [13]. In contrast, the incidence of JHR in adolescents was elevated when compared with other studies. JHR unobserved or unreported by clinicians are potential reasons for this reduced frequency. The delay before antibacterial therapy had a major impact on outcome in adults [8,20]. The need for early initiation of antimicrobial therapy to reduce disease severity remains to be proven in children. However, presumptive treatment based on clinical and epidemiological evidence appears justified while waiting for the laboratory results. Modalities of antibiotics administration to prevent JHR to occur remain to be explored.

Our study suffers from several limitations. First, due to the retrospective design of the study, a significant proportion of

#### References

- Bharti AR, Nally JE, Ricaldi JN, Matthias MA, Diaz MM, et al. (2003) Leptospirosis: a zoonotic disease of global importance. Lancet Infect Dis 3: 757– 771.
- 2. Levett PN (2001) Leptospirosis. Clin Microbiol Rev 14: 296–326.
- Lopes AA, Costa E, Costa YA, Sacramento E, de Oliveira Junior AR, et al. (2004) Comparative study of the in-hospital case-fatality rate of leptospirosis between pediatric and adult patients of different age groups. Rev Inst Med Trop Sao Paulo 46: 19–24.
- Silva HR, Tavares-Neto J, Bina JC, Meyer R (2003) [Leptospiral infection and subclinical presentation among children in Salvador, Bahia]. Rev Soc Bras Med Trop 36: 227–233.
- Karande S, Kulkarni H, Kulkarni M, De A, Varaiya A (2002) Leptospirosis in children in Mumbai slums. Indian J Pediatr 69: 855–858.
- Suarez Hernandez M, Martinez Sanchez R, Posada Fernandez PE, Vidal Garcia I, Bravo Fleites F, et al. (1999) [Human leptospirosis outbreak in the district of Ciego de Avila, Cuba]. Rev Soc Bras Med Trop 32: 13–18.
- Cruz ML, Andrade J, Pereira MM (1994) [Leptospirosis in children in Rio do Janeiro]. Rev Soc Bras Med Trop 27: 5–9.
- Tubiana S, Mikulski M, Becam J, Lacassin F, Lefevre P, et al. (2013) Risk factors and predictors of severe leptospirosis in New Caledonia. PLoS Negl Trop Dis 7: e1991.
- Institut de la Statistique et des Etudes Economiques. Recensement de la population. 2009 Available at: http://www.isee.nc/population/population.html. Accessed 12 May 2013.

patients with untraceable records were secondary excluded of the analysis. Second, leptospiral serogroups were identified in a limited number of cases only. Third, important clinical data were missing, including presence of puberty. Finally, indications for hospitalization may be different in younger children compared to adolescents and mild disease in adolescents may not have been seen because they were not hospitalized.

In New Caledonia, leptospirosis is responsible for a smaller number of hospitalizations in paediatrics due to milder symptomatic forms of the disease. However, leptospirosis remains a public health threat, including in the younger age group which can present with atypical signs and symptoms. Puberty may impact on the severity reported in the older age groups. Physiologic and immunologic factors associated with improved paediatric outcomes require further investigation to provide insights into the pathogenesis of severe leptospirosis.

#### **Supporting Information**

**Checklist S1** STROBE (checklist). (DOC)

#### **Author Contributions**

Conceived and designed the experiments: GG PH ACG EH YP EDO IM. Performed the experiments: GG PH ACG EH YP EDO IM. Analyzed the data: GG PH ACG EH YP EDO IM. Contributed reagents/materials/ analysis tools: GG PH ACG EH YP CG EDO IM. Wrote the paper: GG PH ACG EH YP CG EDO IM.

- Berlioz-Arthaud A, Merien F, Baranton G (2007) [Laboratory based human leptospirosis surveillance in New Caledonia (2001–2005)]. Bull Soc Pathol Exot 100: 133–138.
- Goarant C, Laumond-Barny S, Perez J, Vernel-Pauillac F, Chanteau S, et al. (2009) Outbreak of leptospirosis in New Caledonia: diagnosis issues and burden of disease. Trop Med Int Health 14: 926–929.
- Thai KT, Binh TQ, Giao PT, Phuong HL, Hung le Q, et al. (2006) Seroepidemiology of leptospirosis in southern Vietnamese children. Trop Med Int Health 11: 738–745.
- Libraty DH, Myint KS, Murray CK, Gibbons RV, Mammen MP, et al. (2007) A comparative study of leptospirosis and dengue in Thai children. PLoS Negl Trop Dis 1: e111.
- Marotto PC, Marotto MS, Santos DL, Souza TN, Seguro AC (1997) Outcome of leptospirosis in children. Am J Trop Med Hyg 56: 307–310.
- Lingappa J, Kuffner T, Tappero J, Whitworth W, Mize A, et al. (2004) HLA-DQ6 and ingestion of contaminated water: possible gene-environment interaction in an outbreak of Leptospirosis. Genes Immun 5: 197–202.
- Fialho RN, Martins L, Pinheiro JP, Bettencourt BF, Couto AR, et al. (2009) Role of human leukocyte antigen, killer-cell immunoglobulin-like receptors, and cytokine gene polymorphisms in leptospirosis. Hum Immunol 70: 915–920.
- Thaipadungpanit J, Wuthiekanun V, Chierakul W, Smythe LD, Petkanchanapong W, et al. (2007) A dominant clone of Leptospira interrogans associated with an outbreak of human leptospirosis in Thailand. PLoS Negl Trop Dis 1: e56.

- 18. Adler B, Lo M, Seemann T, Murray GL (2011) Pathogenesis of leptospirosis: the
- Hater B, Eo M, Sternam T, Muray OL (2017) Fathy and Schemes of Repositions. In influence of genomics. Vet Microbiol 153: 73–81.
   Guerrier G, D'Ortenzio E (2013) The jarisch-herxheimer reaction in leptospirosis: a systematic review. PLoS One 8: e59266.
- McBride AJ, Athanazio DA, Reis MG, Ko AI (2005) Leptospirosis. Curr Opin Infect Dis 18: 376–386.